MOSS Minutes

February 3, 2004

ELEMENT ATTENDANCE:

ESDIS SSDO and SSO, MRTT, EDOS, EMOS, ASDC DAAC, OMI SIPS, Averstar, GES DAAC, HIRDLS SIPS, MLS SIPS, TES SIPS

NEXT TELECONS: February 17, 2004 at 2pm EST in Bldg 32 room S241

NEW - DIAL-IN NUMBER:

USA Toll Free Number: 866-581-4760

USA Toll Number: +1-210-280-9695 (international)

PASSCODE: 574686

NEW:  MOSS 4 (5 day test) Data Preparation Schedule (Based on a LRD of no earlier than June 20, 2004):

· February 27: New TES and HIRDLS data for MOSS 4 delivered to ESDIS 

· March 12: Write/verify tapes for EDOS local playback complete 

· March 22-26: MOSS 4 dry run 

· March 30: MOSS 4 TRR 

· April 5-10: Formal MOSS 4 run for the record 

· April 15: Requirements status update complete 

· April 20: MOSS 4 Debrief 

· April 29: Aura SORR Dry Run (date to be confirmed) 

· May 3: Final slides for formal SORR presentation complete (date TBC) 

· May 11: Aura SORR formal presentation (date to be confirmed) 

MOSS 4 Telecon Schedule (all telecons Tuesday at 2:00 PM EST): 

Feb 17, Mar 2, March 16, March 30 (TRR), April 13 (cancelled), April 20 (Debrief)

MEETING HIGHLIGHTS:

Agenda

1. Status of MOSS 4 Data


- GBAD

Karen reported that back in December, FOT ran the simulator to generate new GBAD data for MOSS 4. Carol Elfant saw in that data some values go constant toward the end. She wanted the FOT to re-generate the data but they are having a difficult time trying to get enough consecutive time on the simulator. The simulator is heavily used in Spacecraft Testing. Karen mentioned that she was going to have Carol try to patch the data with MOSS 3 data for the specific problem times. If this works, the data would be checked by DPREP prior to distributing it to the teams. Second option would be to use the data up to the point of the problem, i.e., cut off the data at DOY 108:23:50.  If things have to go this way, all teams confirmed that it would not be a problem for them. Doug of TES mentioned that would prefer to patch the data if possible. The third option (last result) would be to have the teams use OBSIM.

2. Status of ASDC DAAC --> JPL (TES/MISR) network traffic - any issues?

Steve reported that since the AMASS changes in December, he noticed some improvement in MISR flows but discovered in early January that some small files were not making it through (timing out issue). An NCR was written and they are waiting for a fix from Landover. Steve summarized that he would like the following things to happen before the network performance test be conducted:

a. Receive the fix from Landover for the small files timing out

b. Get secure ingest and distribution working in OPS

1. He had some success with F-Secure; F-Secure requires additional information for installation and may require additional support from Landover. Karen informed Steve to let her know if he does not get the level of support he needs.

2. He has used the new TES PDR server (TES SIPS is up and ready to go) and has written ECS internal tracking tickets for configuration changes (Steve to look into the status of the tickets tomorrow --2/4/04).

c. Then perform the network performance testing

3. Status of Secure Ingest and Distribution testing


- ASDC DAAC/TES SIPS 


- GES DAAC/ HIRDLS, MLS, OMI SIPS, ODPS

For ASDC status, see #2 above. 

For GES DAAC, Ed reported that:

· Secure ingest:

·  OPS complete for ODPS, OMI SIPS, HIRDLS SIPS, MLS SIPS

· Secure distribution: 

· TS2 

· Complete for HIRDLS, ODPS, OMI SIPS

· Waiting to hear from Mark Echeverri when testing could start for MLS SIPS

· OPS 

· In process for OMI and HIRDLS SIPS; experiencing problem with Spatial Subscription Server – seeing duplicates but should not; Latha wrote trouble ticket.

4. Status of Engineering Testing with GES DAAC


- MLS 


- HIRDLS (HIR1CAL- EDG)

MLS – Ed reported that the GES DAAC started seeing higher-level products from MLS around 12:06 AM EST Tuesday but they were failing ingest due to directory path not in the PDR (requirement for secure ingest). Ed mentioned that they generally leave polling on but in MLS case they will turn off polling until notified by MLS. Audrey stated that she was not aware that polling stayed on all the time and did not know that GES DAAC was polling the files during their internal test. As a result of this testing, PDR changes are required at the SIPS before testing of the higher-level products could resume. Audrey would notify GES DAAC when they will be ready to start testing and GES DAAC would turn polling back on for MLS.

HIRDLS – Vince mentioned that he would work with the DAAC to test out the HIR1CAL product with EDG. Scott Applebee to verify that the product could be ordered from EDG. Alfreda mentioned that the updated ESDT was delivered to the GES DAAC December 11, 2003, Patch 27 (according to the ECS Science Office weekly spreadsheet).

5. Status of OMI SIPS-ODPS MT#2 re-run; receive new executable?

Marghi reported that the new release of SIPS had some interesting features and they are in touch with the OMI Netherlands to ensure that they clearly understand how the production rules are setup. They plan to have a telecon with the OMI Netherlands tomorrow (2/4/04). Marghi mentioned that all problems would be identified and corrected prior to the MT #2 re-run. She also mentioned that the updated OMI ESDTs (L2s) are in place (in the developers hand) and she has instructed the developers to deliver with the updated ESDTs. Ed at the GES DAAC stated that they also have the ESDTs and Mike Morahan is looking at them. Marghi and Ed would coordinate promotion of the ESDTs when both parties are near ready.  Marghi and Ed are to exchange their list of ESDTs to ensure that they are all the same. Ed mentioned that it would not take long for the GES DAAC to promote these because they are updates to the ones already installed.

6. Status of EDOS-EMOS Secure Transfer


- Test schedule?

Ron reported that it was about two weeks ago when they last interfaced with EOC; a small interface was performed. EDOS send a few files from their high rate and low rate systems to EMOS via sftp and that went okay. Ron is now waiting to hear from EMOS (Hillary/Dan) when a full test could be performed.  Ron hopes to test again before SCIF4, but if cannot, will test with CPT6 data distribution and in MOSS4. He also mentioned that sftp encrypts the data but things should be okay because only small files are involved. For official changes to be applied to OPS, a waiver is required.

7. MOSS 4 Schedule Re-assessment

· Be prepared to provide feedback on the proposed MOSS 4 schedule.

Karen mentioned that management approved the new MOSS 4 dates. The teams accepted the schedule as well.  Rob mentioned that TES would not deliver new data; they would probably need another week. Karen replied that we would work with you and for TES to continue to work on the data. TES is expecting the pre-delivery of their Release 5 on February 9th and it would include everything (i.e., geo-location and performance PGEs).

David mentioned that MLS might deliver some new data. The MLS instrument was not fully turned on in MT #1. David will talk to the L1A engineer and he believes that the data could be delivered by February 20th. He also mentioned that the data would have to be cloned. Karen mentioned that if cloning is required, the sooner the data is delivered the better. After the telecon, David informed Karen via email that MLS will deliver new data for MOSS 4 (data from Thermal Vac testing) but needs to know the preferred format for the delivery. Eric/Bill will contact David with the delivery details.

Ken of HIRDLS will re-deliver new L0 data with noise values included. They hope to delivery by February 20th. 

Karen would re-confirm OMI data plans with Bert. After the telecon, Bill contacted Bert and Bert replied that the current OMI MOSS 4 data plans are still valid. No changes required.

Karen mentioned that FDS do not have to deliver any new products but are requested to deliver the products via the timeline.

8. SCIF 4 Science Support

SCIF4 is currently scheduled to start February 6th at 9am EST and would run thru the weekend. Data would be collected via the Vexcel. The TRR is tomorrow at 3pm. Jerry mentioned that he received three liens for SCIF4 during the Spacecraft Test Team meeting this morning. Jerry read the three liens (1 for HIRDLS, MLS, and TES) and the respected group responded and indicated that their lien would not impact their ability to support SCIF4. After the telecon, the SDPS SCIF 4 TRR slides indicated that the three liens had been closed.

Ed wanted to know when the GES DAAC would begin seeing data. Jerry was to look at the chronology and provide estimated times to the group. After the telecon, Jerry sent out an email with the estimated times and the chronology.

All teams indicated a GO for this weekend test.

9. Additional discussions

CPT 6 - Ron of EDOS mentioned that for CPT6, TES still had some packet length errors and HIRDLS had some packets out of order. TES mentioned that they had received some packets with lengths outside the range that was not expected, both long and short. HIRDLS replied that they have been directed to apply a fix on the ground to address the out of order packets received. After the telecon, Ron (EDOS), Karen and Ken (HIRDLS) conducted a telecon to answer HIRDLS questions regarding 1-second back time jump processing.

John Moses mentioned that he had sent out the Aura Science Operations FORR slides and would appreciate any comments. The final set of charts is due to the project February 17th. He also stated that he would send out another email to confirm that the SORR dates are good for the teams.

Karen mentioned that she has setup a new instant Meet-Me number and that number will be used for all of her future telecons including the OMI SIPS/ODPS bi-weekly test telecons. After the telecon, Karen sent the new Meet-Me number: 

USA Toll Free Number: 866-581-4760

USA Toll Number: +1-210-280-9695 (international)

  
PASSCODE: 574686
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